Author's response to reviews

Title: Up-Beat UK: A programme of research in to the relationship between coronary heart disease and depression in primary care patients.

Authors:

Andre Tylee (a.tylee@iop.kcl.ac.uk)
Mark Ashworth (markashworth@nhs.net)
Elizabeth Barley (e.barley@iop.kcl.ac.uk)
June Brown (june.brown@kcl.ac.uk)
John Chambers (jboydchambers@aol.com)
Anne Farmer (anne.farmer@kcl.ac.uk)
Zoe Fortune (zoe.1.fortune@kcl.ac.uk)
Mark Haddad (m.haddad@iop.kcl.ac.uk)
Rebecca Lawton (rebecca.lawton@kcl.ac.uk)
Anthony Mann (anthony.mann@kcl.ac.uk)
Anita Mehay (anita.mehay@kcl.ac.uk)
Paul McCrone (paul.mccrone@kcl.ac.uk)
Joanna Murray (Joanna.Murray@kcl.ac.uk)
Morven Leese (morven.leese@kcl.ac.uk)
Carmine M Pariante (carmine.pariante@kcl.ac.uk)
Diana Rose (diana.rose@kcl.ac.uk)
Gill Rowlands (rowlang2@lsbu.ac.uk)
Alison Smith (alison.1.smith@kcl.ac.uk)
Paul Walters (p.walters@iop.kcl.ac.uk)

Version: 3 Date: 18 May 2011

Author's response to reviews:

1. authors have been corrected by your colleague after our telephone discussion
2. all method headings are now "methods/design"
3. [17-31] inserted into text about table 1
4. table here text has been removed

thanks